Author's response to reviews

Title: Italian healthcare workers' views on mandatory vaccination

Authors:

Silvio ST Tafuri (s.tafuri@oerpuglia.org)
Domenico DM Martinelli (d.martinelli@oerpuglia.org)
Giovanni GC Caputi (g.caputi@oerpuglia.org)
Annamaria AA Arbore (annamaria.arbore@libero.it)
Cinzia CG Germinario (c.germinario@igiene.uniba.it)
Rosa RP Prato (r.prato@oerpuglia.org)

Version: 2 Date: 27 April 2009

Author's response to reviews: see over
Dear Professor Andrea Bucceri,

We are very delighted for the acceptance of our paper “Italian healthcare workers’ views on mandatory vaccination”.

We revised the manuscript according with the reviewer’s comments and editorial guidelines.

As requested, we shortened the introduction and inserted an acknowledgement section.

All authors have approved the revised manuscript.

With my Best Regards

Rosa Prato
Corresponding Author